\documentclass{revtex4}

\usepackage[english]{babel}
\usepackage[T1]{fontenc}
\usepackage[latin9]{inputenc}
\usepackage{mathpazo}

\begin{document}

% Preprints are sorted by their date

% 2005
\nocite{Carpet-Zink2005b}       % 2005-12-15

% 2006
\nocite{Carpet-Zink2006c}       % 2006-06-29
\nocite{Carpet-Sperhake2006b}   % 2006-07-23
\nocite{Carpet-Zink2006b}       % 2006-07-23
\nocite{Carpet-Baiotti2006c}    % 2006-09-04
\nocite{Carpet-Sperhake2006c}   % 2006-09-04
\nocite{CarpetResult-Font2006a} % 2006-09-04

% 2007
\nocite{CarpetResult-Baker2007a} % 2007-04-23
\nocite{Carpet-Schnetter2007a}  % 2007-07-11 (book chapter)
\nocite{Carpet-Stark2007a}      % 2007-09-09

% 2008

% 2009
\nocite{CarpetResult-Hannam2008a} % 2009-01-20
\nocite{CarpetResult-Nakano2008a} % 2009-01-24



\bibliographystyle{apsrev-titles-manyauthors}
\bibliography{publications}

\end{document}
